Author's response to reviews

Title: Impact of nephrolithiasis on kidney function

Authors:

Vaka K Sigurjonsdottir (vakaks@gmail.com)
Hrafnhildur L Runolfsdottir (hilda_linnet@hotmail.com)
Olafur S Indridason (olasi@lsh.is)
Runolfur Palsson (runolfur@lsh.is)
Vidar O Edvardsson (vidare@lsh.is)

Version: 4 Date: 22 June 2015

Author's response to reviews: see over
June 22, 2015

BMC Nephrology
BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Re: MS: 1199529082159519 – the fully revised manuscript, entitled “Impact of nephrolithiasis on renal function”, authored by Vaka K Sigurjonsdottir, Hrafnhildur L Runolfsdottir, Olafur S Indridason, Runolfur Palsson and Vidar O Edvardsson.”

To whom it may concern:

Enclosed is the final version of the manuscript, entitled “Impact of nephrolithiasis on renal function”. As requested by the editors, the name of the National Bioethics Committee that gave approval for our study is now documented at the end of the first paragraph of the Methods section (The study was approved by The National Bioethics Committee of Iceland (VSN 03-002-V3) and the Icelandic Data Protection Authority). No further changes were requested by the Editorial staff.

We greatly appreciate the opportunity to publish our work in BMC Nephrology.

Sincerely,

Vidar O. Edvardsson, MD
Children’s Medical Center, 21-D
Landspitali – The National University Hospital of Iceland
Hringbraut
101 Reykjavik
Iceland
E-mail: vidare@landspitali.is